Author's response to reviews

Title: Design of the Prevention of Adult Caries Study (PACS): A randomized clinical trial assessing the effect of a chlorhexidine dental coating for the prevention of adult caries

Authors:

William L Vollmer (william.vollmer@kpchr.org)
Athena S Papas (athena.papas@tufts.edu)
James D Bader (jim_bader@unc.edu)
Gerardo Maupome (gmaupome@iupui.edu)
Christina M Gullion (christina.quillion@kpchr.org)
Jack F Hollis (jack.hollis@kpchr.org)
John J Snyder (John.J.Snyder@kp.org)
Jeffrey L Fellows (jeffrey.fellows@kpchr.org)
Reesa L Laws (reesa.laws@kpchr.org)
PACS Collaborative Research Group (william.vollmer@kpchr.org)

Version: 3 Date: 26 August 2010

Author's response to reviews: see over
August 25, 2010

*BMC Oral Health Journal*
Melissa Norton, MD
Editor-in-Chief

Dear Dr. Norton,

We are resubmitting our manuscript, Design of the Prevention of Adult Caries Study (PACS): A randomized clinical trial assessing the effect of a chlorhexidine dental coating for the prevention of adult caries,” with the following changes:

- We have removed B. Alexander White as an author.
- We have added a paragraph that explicitly states each author’s contributions.
- We have added a functioning link to reference #32.
- We have replaced shading in our tables with boldface type.
- We have changed the vertical text in Table 6.

Please contact me if you have any further questions about our manuscript.

Sincerely,

William M. Vollmer, PhD
Senior Investigator
Center for Health Research
Kaiser Permanente Northwest